Author's response to reviews

Title: Home Exposure to Arabian Incense (Bakhour) and Asthma Symptoms in Children: A Community Survey in Two Regions in Oman

Authors:

Omar A Al-Rawas (orawas@squ.edu.om)
Abdullah A Al-Maniri (almaniri@gmail.com)
Bazdawi M Al-Riyami (bazdawi@squ.edu.om)

Version: 5 Date: 25 April 2009

Author's response to reviews: see over
To: Mark Todd  
Assistant Editor,  
BMC-series Editor,  

Dear Sir,  

RE: MS: 5046163272194702 (Home Exposure to Arabian Incense (Bakhour) and Asthma Symptoms in Children: A Community Survey in Two Regions in Oman)  

Thank you for the provisional acceptance of our above manuscript. We have reviewed the comment of the reviewers and were pleased to note that there were only two minor corrections suggested by Referee # 4: Dr. Giorgio Bedongi which we have corrected; namely:  

• "P8 L4 Please add SD of age for South Sharqiyah and Muscat"  
• "P8 L5 Please add % after 54.3 and 51.3"  

We also removed the authors' qualifications from the Title Page and included the e-mail addresses for all authors on the Title Page.  

Looking forwards to hearing from you  

Regards,  

Dr. Omar Al-Rawas  
Corresponding Author